Author's response to reviews

Title: Interference with work in fibromyalgia - effect of treatment with pregabalin and relation to pain response

Authors:

  Sebastian Straube (sebastian.straube@googlemail.com)
  R Andrew Moore (andrew.moore@nda.ox.ac.uk)
  Jocelyn Paine (popx@i-paine.org)
  Sheena Derry (sheena.derry@pru.ox.ac.uk)
  Ceri J Phillips (C.J.Phillips@swansea.ac.uk)
  Ernst Hallier (ehallie@gwdg.de)
  Henry J McQuay (henrymcquay@gmail.com)

Version: 4 Date: 1 June 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team

Thank you for your comments. I’m uploading a revised version of the manuscript text and figures, with changes according to your requests. A point-by-point response follows:

1. “Major revisions (we require the author to make these changes). Structure: this manuscript does not have the correct structure. Please reformat it to use these headings:
   Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any);
   Competing interests; Authors' contributions; Authors’ information (optional);
   Acknowledgements (optional); References”

Authors’ response: I have reformatted the manuscript to conform to the suggested structure. I have added ‘Conclusions’ and ‘Abbreviations’ sections.

2. Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above). Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:

   http://www.biomedcentral.com/info/ifora/figures”

Authors’ response: I have cropped the figures even more, white space around the images is now minimal.
3. “Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.”

Authors’ response: Done.

Please let me know if I can be of further assistance.

With best wishes

Sebastian Straube